ADMINISTRATIVE DETAILS

Reading:

ffl csc324 Course Information sheet.

You must read this.

Contact Hours Me: Office hours:

See the info sheet.

Tutorials:

ffl Tutorials will usually involve examples of lecture material, but may cover new material also.

ffl See the web page for your tutorial.

You may not switch tutorials without a good reason and my permission.

Textbooks etc.

Required: (There is no official textbook) ffl a Student’s Guide to CDF,

Fall 1999 edition.

Recommended References:

ffl Sethi, Programming Languages: Concepts and Constructs, Addison-Wesley, 1996.

ffl Sebesta Concepts of Programming Languages, Addison-Wesley 1999.

ffl Pratt and Zelkowitz, Programming Languages: Design and Implementation, 3rd edition, Prentice Hall, 1999.

ffl Lisp reference: Wilensky, Common Lispcraft, Norton & Company, 1986.

ffl Prolog reference: Clocksin and Mellish, Programming in Prolog, Springer-Verlag, 1994.

Assignments and Tests

Item Date Weight Comment Quiz (week 03) 2% Assnt 1 (week 04) 5% Lisp basics Quiz (week 05) 2% Quiz (week 06) 2% Assnt 2 (week 07) 10% Lisp Mid-term (week 08) 15% Assnt 3 (week 09) 5% Prolog basics Quiz (week 10) 2% Quiz (week 11) 2% Assnt 4 (week 12) 10% Prolog Assnt 5 (week 13) 5% paper assnt Final exam period 40% three hour

All assignments are to be done individually.

A5 is due on a Thursday, with no grace days allowed.

You need 40% on the final to pass the course. 4

Working on our computers

Lab:

ffl The lab for this course is CDF, located in the Engineering Annex.

ffl Door combination:

ffl See A Student’s Guide to CDF for more information about the lab.

Accounts:

ffl You account name is a324xxxx, where xxxx is usually the first four letters of your surname.

ffl Your initial password is your student number. Change it.

Print quota:

ffl Your quota is 300 sides. See A Student’s Guide to CDF for more information.

Working from home

You may use your own computer to do your assignments for this course, if you have software for writing Common Lisp and Prolog.

Notes:

ffl Your code must run correctly on cdf.

ffl We can’t help you if you have problems with your computer. Such problems will not merit an extension on an assignment.

ffl You will have to use the cdf lab at least once per assignment, in order to hand in your code electronically.

Plagiarism

The work you submit must be your own: both words and ideas.

All cases of suspected plagiarism are reported to the Chair of Computer Science and/or the Dean of Arts and Science.

Both presenting someone else’s work as your own, and letting someone else use your work are offenses.

You may discuss general approaches to assignments with others, but

ffl you must not have any written notes from the discussion, and

ffl you must not show your solution (even in draft form) to anyone else or look at theirs.

Late Policy

Assignments are due in tutorial.

Late assignments must be handed in to the 324 drop box.

A “grace days” policy will be used:

ffl You have 2 grace days to use. ffl 4:00 p.m. the next day = one grace day. ffl 4:00 a.m. the following day = 2 grace days.

ffl Once you have exhausted your grace days, you cannot hand in anything late for marks.

Grace days are for emergencies, not for being busy.